Improving quality of care and outcome at very preterm birth: the Preterm Birth research programme, including the Cord pilot RCT

Lelia Duley, 1* Jon Dorling, 2 Susan Ayers, 3 Sandy Oliver, 4 Charles William Yoxall, 5 Andrew Weeks, 6 Chris Megone, 7 Sam Oddie, 8 Gill Gyte, 9 Zoe Chivers, 10 Jim Thornton, 2 David Field, 11 Alexandra Sawyer 12 and William McGuire 8 on behalf of the Preterm Birth Programme Collaborative Group

- ¹Nottingham Clinical Trials Unit, University of Nottingham, Nottingham, UK
- ²Department of Child Health, Obstetrics and Gynaecology, University of Nottingham, Nottingham, UK
- ³Centre for Maternal and Child Health Research, School of Health Sciences, City, University of London, London, UK
- ⁴Social Science Research Unit and EPPI-Centre, Institute of Education, University of London, London, UK
- ⁵Neonatal Unit, Liverpool Women's Hospital, Liverpool, UK
- ⁶University of Liverpool and Liverpool Women's Hospital, Members of Liverpool Health Partners, UK
- ⁷Inter Disciplinary Ethics Applied, University of Leeds, Leeds, UK
- ⁸Centre for Reviews and Dissemination, University of York, York, UK
- ⁹National Childbirth Trust, London, UK
- ¹⁰Bliss, London, UK
- ¹¹Department of Health Sciences, University of Leicester, Leicester, UK
- ¹²School of Health Sciences, University of Brighton, Brighton, UK

Declared competing interests of authors: Lelia Duley reports National Institute for Health Research (NIHR) Clinical Trials Unit support funding. Jon Dorling reports grants from NIHR during the conduct of the study, grants from NIHR and grants from Nutrinia Ltd (Ramat Gran, Israel) outside the submitted work. He also reports membership to the Health Technology Assessment (HTA) Efficient Study Designs 2015–16, HTA General Board 2016–19, HTA Maternal, Neonatal and Child Health (MNCH) Methods Group 2014–18, HTA MNCH Panel 2013–18, and HTA Post Board Funding Teleconference 2015–18. Andrew Weeks reports grants from Liverpool Women's Hospital Charity during the conduct of the study. Gill Gyte reports personal fees from Cochrane Pregnancy and Childbirth Group outside the submitted work. Jim Thornton reports membership of HTA Clinical Trials Board 2010–15, HTA and EME Editorial Board 2016–18, Rapid Trials and Add on Studies 2012, CPR Decision-Making Committee. David Field reports HTA Commissioning Board 2013–2018, HTA and Efficacy and Mechanism Evaluation (EME) Editorial Board 2012–18. William McGuire reports grants from NIHR during the conduct of the study, and membership of the HTA Commissioning Board 2013–present and the HTA and EME Editorial Board 2012–present.

^{*}Corresponding author lelia.duley@nottingham.ac.uk

Published September 2019 DOI: 10.3310/pgfar07080

Plain English summary

The Preterm Birth research programme

Programme Grants for Applied Research 2019; Vol. 7: No. 8 DOI: 10.3310/pgfar07080

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Babies born before 32 weeks' gestation (very preterm) may have poor health or they may not survive. Our research focused on care at the birth, with close co-operation between parent representatives, clinicians and researchers.

We interviewed parents who described their experience of very preterm birth as a 'rollercoaster of emotions', and said that touching their baby helped bonding. About one-third of parents see and touch their baby for the first time on the neonatal unit. Parents were mostly positive about care during the birth, but some women felt that they had not been listened to or had not been believed; in addition, some fathers felt excluded. Parents found it helpful when staff were calm and took control. We developed a questionnaire to measure parents' satisfaction with care during preterm birth.

We designed a small trolley so that immediate care for the baby could be beside their mother, and also showed that this care could be done using existing equipment. Parents and clinicians were largely positive about this care.

We do not know when is best to clamp the umbilical cord at birth. Our randomised trial at very preterm birth found that cord clamping after ≥ 2 minutes and initial care for the baby with the cord intact may be beneficial, compared with immediate clamping and care after clamping. A brief process for offering participation in the trial (consent) when birth was imminent, with a detail explanation later, was acceptable to women and clinicians. We will pool our data with other trials to help find out the best way of managing cord clamping at preterm birth.

Through involving parents, the public and health-care professionals, we identified 104 unanswered research questions around care for preterm birth, and ranked the top 30.

Our work has improved understanding of parents' experiences, provided research ideas and developed a consent process and ways of caring for babies at birth beside their mother.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-0609-10107. The contractual start date was in March 2011. The final report began editorial review in April 2018 and was accepted for publication in June 2019. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, NETSCC, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PGfAR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Duley et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Honorary Professor, University of Manchester, and Senior Clinical Researcher and Associate Professor, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk